\begin{table}%Tc1
\caption{S and Zn abundance measurements in DLAs.}
\label{tab:C.1}
\small%\centerline
{
\begin{tabular}{lrcccccr}
\hline\hline
\noalign{\smallskip}
  \multicolumn{1}{c}{QSO}
& \multicolumn{1}{c}{$z_{\rm abs}$} 
& \multicolumn{1}{c}{log $N$\/(H~I)}
& \multicolumn{1}{c}{log $N$\/(S~II)}
& \multicolumn{1}{c}{log $N$\/(Zn~II)}
& \multicolumn{1}{c}{[Zn/H]$^{a}$}
& \multicolumn{1}{c}{[S/Zn]$^{b}$}
& ~~Ref.$^{c}$\\
    & 
& \multicolumn{1}{c}{(cm$^{-2}$)}
& \multicolumn{1}{c}{(cm$^{-2}$)}
& \multicolumn{1}{c}{(cm$^{-2}$)}
& 
&
& \\ 
\hline
Q0000$-$2620        & 3.3901   & $21.41 \pm 0.08$ & $14.70 \pm 0.03$  & $12.01 \pm 0.05$ & $-2.01$ &   $+0.14$ &   10 \\
Q0013$-$004         & 1.97296  & $20.83 \pm 0.05$ & $15.28 \pm 0.03$  & $12.74 \pm 0.04$ & $-0.70$ &   $-0.01$ &   11 \\
Q0100$+$130         & 2.30903  & $21.37 \pm 0.08$ & $15.09 \pm 0.06$  & $12.47 \pm 0.01$ & $-1.51$ &   $+0.07$ &   2  \\
Q0201$+$365         & 2.4628   & $20.38         $ & $15.29 \pm 0.01$  & $12.76 \pm 0.05$ & $-0.23$ &   $-0.02$ &   14, 12 \\  
Q0407$-$4410        & 2.5505   & $21.13 \pm 0.10$ & $14.82 \pm 0.06$  & $12.44 \pm 0.05$ & $-1.30$ &   $-0.17$ &   8 \\
Q0407$-$4410        & 2.5950   & $21.09 \pm 0.10$ & $15.19 \pm 0.05$  & $12.68 \pm 0.02$ & $-1.02$ &   $-0.04$ &   8 \\
B0528$-$250         & 2.8120   & $21.11 \pm 0.04$ & $15.56 \pm 0.02$  & $13.27 \pm 0.03$ & $-0.45$ &   $-0.26$ &   1  \\
Q0551$-$366         & 1.96221  & $20.50 \pm 0.08$ & $15.38 \pm 0.11$  & $13.02 \pm 0.05$ & $-0.09$ &   $-0.19$ &   6 \\
FJ081240.6$+$320808 & 2.6263   & $21.35         $ & $15.63 \pm 0.08$  & $13.15 \pm 0.02$ & $-0.81$ &   $-0.07$ &   13  \\
0841$+$1256         & 2.37452  & $20.99 \pm 0.08$ & $14.69 \pm 0.05$  & $12.10 \pm 0.02$ & $-1.50$ &   $+0.04$ &   3  \\
0841$+$1256         & 2.47621  & $20.78 \pm 0.08$ & $14.48 \pm 0.10$  & $11.69 \pm 0.10$ & $-1.70$ &   $+0.24$ &   3 \\
SDSS1116$+$4118A    & 2.9422   & $20.28 \pm 0.05$ & $15.01 \pm 0.10$  & $12.40 \pm 0.33$ & $-0.49$ &   $+0.06$ &   5 \\
LBQS~1210$+$1731    & 1.89177  & $20.63 \pm 0.08$ & $14.96 \pm 0.02$  & $12.40 \pm 0.05$ & $-0.84$ &   $+0.01$ &   3 \\
Q1331$+$170         & 1.77637  & $21.14 \pm 0.08$ & $15.08 \pm 0.11$  & $12.54 \pm 0.02$ & $-1.21$ &   $-0.01$ &   2  \\
HE2243$-$6031       & 2.33000  & $20.67 \pm 0.02$ & $14.88 \pm 0.01$  & $12.22 \pm 0.03$ & $-1.06$ &   $+0.11$ &   7 \\
B2314$-$409         & 1.8573   & $20.90 \pm 0.10$ & $15.10 \pm 0.05$  & $12.52 \pm 0.03$ & $-0.99$ &   $+0.03$ &   4 \\
LBQS~2230$+$0232    & 1.86359  & $20.83 \pm 0.05$ & $15.29 \pm 0.06$  & $12.72 \pm 0.05$ & $-0.72$ &   $+0.02$ &   3  \\
Q2231$-$002         & 2.06616  & $20.53 \pm 0.08$ & $15.10 \pm 0.15$  & $12.30 \pm 0.05$ & $-0.84$ &   $+0.25$ &   2  \\
Q2343$+$125         & 2.4313   & $20.35 \pm 0.05$ & $14.71 \pm 0.08$  & $12.45 \pm 0.06$ & $-0.51$ &   $-0.29$ &   9 \\
Q2343--BX415        & 2.5739   & $20.98 \pm 0.05$ & $15.38 \pm 0.03$  & $12.90 \pm 0.06$ & $-0.69$ &   $-0.07$ &   15 \\
\hline
\end{tabular}}
\smallskip
 $^{a}$
[$\log$ (Zn/H)$_{\rm DLA} - \log$ (Zn/H)$_{\odot}$],
where $\log$ (Zn/H)$_{\odot} = -7.39$ (Asplund et~al. 2005).\\
$^{b}$
[$\log$ (S/Zn)$_{\rm DLA} - \log$ (S/Zn)$_{\odot}$],
where $\log$ (S/Zn)$_{\odot} = 2.55$ (Asplund et~al. 2005).\\
$^{c}$
References --
1:~Cent\'{u}rion et~al. (\cite{centurion03});
2:~Dessauges-Zavadsky et~al. (\cite{dessauges04}):
3:~Dessauges-Zavadsky et~al. (\cite{dessauges06},\cite{dessauges07});
4:~Ellison \& Lopez (\cite{ellison01});
5:~Ellison et~al. (\cite{ellison01});
6:~Ledoux et~al. (\cite{ledoux02});
7:~Lopez et~al. (\cite{lopez02});
8:~Lopez \& Ellison (\cite{lopez03});
9:~Lu et~al. (\cite{lu98});
10:~Molaro et~al. (\cite{molaro00}); 
11:~Petitjean et~al. (\cite{petitjean02});
12:~Prochaska et~al. (\cite{prochaska07});
13:~Prochaska et~al. (\cite{prochaska03});
14:~Prochaska \& Wolfe (\cite{prochaska99});
15:~Rix et~al. (\cite{rix07}).
\end{table}